Author's response to reviews

Title: A preliminary analysis of the effect of the new rural cooperative medical system on inpatient care at a county hospital.

Authors:

Chiyu Ye (jumpmanyes@163.com)
Shengnan Duan (21018240@zju.edu.cn)
Yuan Wu (21018232@zju.edu.cn)
Huimei Hu (huhuimei@zju.edu.cn)
Xiaofang Liu (21118250@zju.edu.cn)
Hua You (youthua98@163.com)
Linghao Wang (21218252@zju.edu.cn)
Hengjin Dong (donghj@zju.edu.cn)

Version: 2 Date: 13 February 2013

Author's response to reviews: see over
Dear Sir/Madam,

RE: Manuscript submission “A preliminary analysis of the effect of the new rural cooperative medical system on inpatient care at a county hospital”

Please accept my greetings from Hangzhou, China. I hereby submit a manuscript titled “A preliminary analysis of the effect of the new rural cooperative medical system on inpatient care at a county hospital” This article aims to evaluate the effect of the new rural cooperative medical system (NCMS) on inpatient care utilization during 2003 to 2012 at one of county hospitals.

We hope that you will find this manuscript of interest for the readers of your journal because NCMS has improved equity of health utilization of Chinese and made big contribution for universal coverage.

This manuscript is an original unpublished work and is not being submitted for publication elsewhere at the same time.

Hoping to hear from you soon!

Yours truly,

Hengjin Dong, PhD